Answering Reviewers

Reviewer #1:
Scientific Quality: Grade B (Very good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Minor revision
Specific Comments to Authors: Abdominal bronchogenic cysts are rarely documented. In this study, the authors report a rare case of a bronchogenic cyst discovered in the abdominal cavity. The case is very interesting. The Case presentation section should be re-arranged according to the journal’s guideline. Conclusion section in the main text is too long, please move the content about the discussion to the “Discussion” section. References should be edited.

Answering Reviewer 1:
We have revised all the related parts as the first reviewer suggested. The Case Presentation was rearranged according to the journal’s guideline. The “conclusion section” was condensed into a more brief one; the references were also reedited in line with the format of references.

Reviewer #2:
Scientific Quality: Grade B (Very good)
Language Quality: Grade B (Minor language polishing)
Conclusion: Minor revision
Specific Comments to Authors: This is an interesting case report of abdominal bronchogenic cyst. The manuscript is well written, and the case is well described. After a minor editing, it can be accepted for publication.

Answering Reviewer 2:
We have carefully polished and re-edited this case report according to the “Format for Manuscript Revision Case Report”.

We would like to express our sincere gratitude to the two reviewers’ revision suggestions.